Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/cemconcomp



# Piezoelectric composite cements: Towards the development of self-powered and self-diagnostic materials

Check for updates

Daniel A. Triana-Camacho<sup>a</sup>, Jorge H. Quintero-Orozco<sup>a</sup>, Enrique Mejía-Ospino<sup>b</sup>, Germán Castillo-López<sup>c</sup>, Enrique García-Macías<sup>d,\*</sup>

a Ciencia de Materiales Biológicos y Semiconductores (CIMBIOS), Escuela de Física, Universidad Industrial de Santander, Cra 27 Calle 9, Bucaramanga, Colombia

<sup>b</sup> Laboratorio de Espectroscopía Atómica y Molecular, Escuela de Química, Universidad Industrial de Santander, Bucaramanga, Colombia

<sup>c</sup> Escuela de Ingenierías Industriales, Departamento de Ingeniería Civil de Materiales y Fabricación, Calle Dr. Ortiz Ramos s/n, 29071, Málaga, Spain

# ARTICLE INFO

Keywords: Cement Cyclic voltammetry Composite Graphene oxide Piezoelectricity Piezoresistivity Self-diagnostic material

# ABSTRACT

Piezoresistivity is the most commonly used sensing principle in cement-based smart composites for strainmonitoring applications. Nonetheless, the need for external electric power to conduct electrical resistivity measurements restricts the scalability of this technology, especially when implemented in remote structures. To address this issue, this manuscript thoroughly analyses the piezoelectric properties of cement composites doped with reduced graphene oxide (rGO) and evaluates their potential as self-powered strain sensors. To do so, a comprehensive methodology involving voltammetry measurements, open circuit potential determination, and uniaxial compression testing is developed to determine the piezoelectric coefficients of charge  $d_{33}$  and voltage  $g_{33}$ . Furthermore, a novel circuital model for signal processing of the electromechanical response is developed and experimentally validated in terms of time series of output voltage, resistance, and the generated electric power. The developed methodology is applied to laboratory samples manufactured following two different filler dispersion methods. The presented results evidence that samples prepared by ultrasonic cleaner dispersion achieve optimal properties, with a piezoelectric charge coefficient of  $1122.28 \pm 246.67$  pC/N, about 47 times greater than previously reported composites in the literature. Unlike piezoresistive cement-based composites, a remarkable nonlinear correlation between the fractional change in the intrinsic resistance of the material and the applied mechanical strain has been observed. Instead, a considerable linearity ( $R^2 = 0.96$ ) between the externally applied mechanical strain and the generated (piezoelectric) electric power has been found, which suggests the great potential of the latter for conducting off-the-grid strain monitoring applications.

# 1. Introduction

Concrete constitutes the most widely-used construction material worldwide with a global production of  $\approx 26$  Gt/year [1]. The production of Portland cement, the main component of concrete, is estimated at 4.4 Gt/year and prospects indicate that this figure may exceed 5.5 Gt/year by 2050 [2]. The rapid development of the concrete industry followed the economic boom after World War II, and a large section of the built infrastructure was constructed during the second half of the 20<sup>th</sup> century. Nonetheless, the life-span of reinforced concrete structures is typically of 50–100 years or even less [3], which implies that a broad section of the built stock has reached or surpassed its life expectancy. Evidence of the formidable challenge posed by the management of ageing infrastructure are some of the most recent catastrophic collapses such as the Morandi Bridge in 2018 (Genoa,

Italy) [4] of the Fern Hollow Bridge in 2022 (Pittsburg, US). This circumstance has fostered the development of new infrastructure maintenance plans and standards (see e.g. [5,6]), as well as considerable funding efforts devoted to R&D actions in the realm of Structural Health Monitoring (SHM). In this context, the development of innovative micro- and nano-engineered multifunctional materials has rendered broad new possibilities in the realm of SHM. In particular, self-sensing cementitious composites (SSCCs) offer clear advantages with respect to traditional monitoring technologies (e.g. strain gauges, optic sensors, piezoelectric ceramic, accelerometers, etc.). These innovative materials can be cast in the shape of load-bearing sensors offering high compatibility with the host structure (both through coatings or embeddable sensors, as well as distributed sensing solutions) and similar durability to standard concrete [7–9]. Most research efforts have been devoted

\* Corresponding author. *E-mail addresses:* dantrica@saber.uis.edu.co (D.A. Triana-Camacho), jhquinte@uis.edu.co (J.H. Quintero-Orozco), enriquegm@ugr.es (E. García-Macías).

https://doi.org/10.1016/j.cemconcomp.2023.105063

Received 24 November 2022; Received in revised form 1 March 2023; Accepted 3 April 2023 Available online 3 April 2023

<sup>&</sup>lt;sup>d</sup> Department of Structural Mechanics and Hydraulic Engineering, University of Granada, Av. Fuentenueva sn, 18002 Granada, Spain

<sup>0958-9465/© 2023</sup> The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

to the development of piezoresistive cement-based composites, whose self-sensing ability stems from the dependency of their electrical conductivity upon mechanical deformations. Hence, these materials allow getting information about the structural integrity and detecting damage through electrical resistivity measurements [10]. The last decade has witnessed broad engineering applications of SSCCs [11], such as strain sensing [12], damage detection and localization [13], structural modal identification [14], monitoring of train loads [15], weigh-in-motion systems [16], and human motion detection [17], to mention a few. Nonetheless, despite the obvious motivation of these new materials, their technological transfer to routine engineering practice is anecdotal, and their applicability remains virtually exclusive to laboratory environments.

Carbon-based conductive nano-fillers are becoming increasingly popular for the development of SSCCs, including carbon fibres (CFs), carbon black (CB), carbon nanotubes (CNTs), graphene (G) and derivatives such as graphene nano-platelets (GnPs), graphene oxide (GO), or rGO [18]. When dispersed into a cementitious material, the resulting composite gains enhanced electrical conductivity, which may be of several orders of magnitude higher than that of the pristine matrix. For instance, Qureshi and Panesar [19] reported a reduction of 18% in the 28-day electrical resistance of cement when adding rGO at a small concentration of 0.16 wt%. The electrical transport properties of SSCCs are typically explained through the percolation theory accounting for two different conduction mechanisms, namely tunnelling and conductive networking [20]. The percolation threshold indicates the critical filler volume fraction at which the doping particles touch one another, originating continuous conductive paths with the subsequent drastic increase of the overall conductivity. Below percolation, the transfer of electrons takes place through the (dielectric) matrix and the potential barrier between non-contacted fillers in virtue of a quantum tunnelling mechanism. Above percolation, both mechanisms act simultaneously, although conductive networking becomes dominant. In this light, the self-sensing property of SSCCs can be understood as the alteration of these mechanisms driven by mechanical strains.

The origin of the strain self-sensitivity of SSCCs has been ascribed in the literature to three different mechanisms [21]: (i) piezoresistivity, (ii) piezopermittivity, and (iii) piezoelectricity. Piezoresistivity constitutes the most extensively investigated mechanism. It is defined as the one-way coupling between the material electrical resistivity and mechanical strains [22], that is to say, mechanical loads affect the electric field but not the opposite. Jang et al. [23] investigated the electrical and piezoresistive properties of cement-based sensors doped with multi-walled CNTs (MWCNTs) and CB, respectively, exposed to various temperatures. Their results reported gauge factors (GFs) in the range  $\approx 18-50$  for temperature values between 25 and 400 °C and MWCNTs contents of 0.2, 0.5, and 1 wt%. Similarly, Dong and co-authors [24] investigated the piezoresistive properties of cement doped with 3 wt% CB under various temperature and water content conditions. Their findings revealed that the piezoresistive properties were not significantly affected by thermal variations in the range of -20 to 100 °C, once the heat gradients between the sensors and the working environment were removed. Conversely, the water content had a more significant impact, with optimal sensitivities achieved at a content of around 8%, resulting in a GF of 488. Xu and Zhang [25] and Guo et al. [26] reported GFs in the ranges 0.08-7.78 (0.67-2.00 wt%) and 3-16 (0.05-0.15 wt%) for cement doped with GnP and GO, respectively. Both experimental and theoretical evidence in the literature have shown that increasing compressive strains lead to decreases in the electrical resistivity of the composite and vice-versa for tensile strains. Other recent studies have focused on analysing the piezoresistive properties of rGO/cement composites under conditions of sulphuric acid attack [27] and dynamic mechanical excitation [28]. Piezopermittivity describes the effect of mechanical strains upon the dielectric behaviour of a material. In general, the capacitance of cementitious materials stems from the movement of charge carriers under

an electric field. Specifically, electric fields induce the ions in the medium to be repelled causing a dipolar effect, thus resulting in polarization. When subjected to a constant potential difference, polarization in cement-based materials manifests as electric current decreases over time as a result of capacitive effects [29,30]. With respect to the relevance of piezocapacitive effects in the electromechanical response of SSCCs, disparate conclusions have been reported in the literature. For instance, Han et al. [31] found no significant variation in the internal capacitance of cement composites doped with low filler contents of MWCNTs (0.1 and 0.5 wt%) under low compressive loadings ( $\leq 6$  MPa). Conversely, Hou and co-authors [32] reported comparable GFs in terms of internal resistance and capacitance. Those authors investigated the electrical properties of cement doped with steel fibres at concentrations between 0.5 and 2 vol.% through DC and AC resistivity measurements and large compressive loads (0-25 MPa). Their results evidenced that capacitance-based measurements provide noisier signals and slightly lower strain sensitivities compared to resistance-based measurements. Finally, piezoelectricity relates the property of certain materials to produce an external voltage when a mechanical load is exerted on it [33]. Unlike piezoresistivity, piezoelectricity represents a two-way coupling where mechanical strains affect the electric field and vice-versa. The intrinsic piezoelectric properties of cement hydrates without functional fillers have been long recognized in the literature [34,35]. It is believed that the appearance of permanent dipoles in cement-based materials originating piezoelectricity stems from the interaction of mobile charge carriers (e.g. the ions in the pore solution) with the atoms or ions in the solid phase, particularly those associated with certain micro-structural elements such as the diffuse interfaces in cement paste. However, typical piezoelectric coefficients of hardened cement are in the order of 1.0E-8 A/MPa, which is too small to be exploited for monitoring purposes. For this reason, researchers have strived to increase the piezoelectricity of cementitious materials, primarily through the addition of piezoelectric functional fillers such as lead zirconium titanate (PZT) or polyvinylidene fluoride (PVDF) (see e.g. [36,37]). In this light, the incorporation of conductive carbon-based nanofillers has been mainly investigated for the purpose of reducing the polarization voltage of the resulting three-phase composite and enhancing its effectiveness. For instance, Gong and co-authors [38] reported that the addition of 1.0 vol.% CB to PZT/cement allowed achieving an efficient polarization at 40 kV/cm and room temperature, attaining an effective piezoelectric strain constant  $d_{33}$  of 28.5 pC/N. Similar experiences in the literature concerning the use of CNTs, CB or GnPs (refer to e.g. [39-41]) report the achievement of considerably high piezoelectric strain constants  $d_{33}$ within the range 13-122 pC/N. Overall, SSCCs exploiting piezoelectricity require no power supply (neither DC nor AC) to conduct electrical measurements, which represents a clear practical advantage with respect to piezoresistive SSCCs. This feature may effectively boost the technological transfer of SSCCs towards their wide-scale implementation, since the provision of power supply to civil engineering structures at a regional level is simply infeasible. Nonetheless, the low toughness and the health and environmental concerns related to piezoelectric ceramics represent a major limitation for their practical use to develop load-bearing SSCC sensors. Alternatively, recent research works have proposed the use of DC electric fields to polarize the ions inside the fresh cement paste while curing, so enhancing its intrinsic piezoelectric properties without piezoceramics (see e.g. [42,43]). In this line, a noteworthy contribution was made by Al-Qaralleh [44] who reported considerable enhancements in the piezoelectric properties of cements polarized while curing at a DC potential of 5 V. Specifically, that work reported a piezoelectric voltage constant g<sub>33</sub> of 9.7E-5 mV m/N, a value considerably superior to that of untreated cement (3.7E–5 mVm/N). Nonetheless, whilst promising, the use of polarization techniques in the fabrication of SSCCs may limit its scalability. From the literature review above, it is clear that the use of carbon-based functional fillers to directly enhance the intrinsic piezoelectric properties of cementitious materials without the addition of piezoceramic inclusions remains scarcely explored.

Experimental research has been accompanied by considerable efforts devoted to the development of theoretical models capable of interpreting the underlying physical phenomena governing SSCCs. Overall, theoretical models reported in the literature comprise mean-field homogenization (MFH) techniques, numerical simulation, and lumped circuital models. Mean-field homogenization offers a powerful analytical or semi-analytical framework to estimate the effective physical properties of SSCCs by exploiting the statistical averaging of the contribution of the constituent phases within a representative volume element (RVE) [45]. In this regard, it is worth noting the work by Takeda et al. [46] who presented a micromechanics model of the electrical conductivity of CNTs/polymer composites, which allowed to distinguish the contribution of tunnelling-type contacts and conductive networking, as well as to simulate percolation and filler waviness. Another prominent contribution was made by Seidel and Lagoudas [47] who proposed an Eshelby-Mori-Tanaka model for the study of the individual influence of the conductive mechanisms on the overall electrical conductivity of CNTs/polymer composites. Tallman and Wang [48] extended the formulation in [46] to simulate the piezoresistive properties of CNTs-based composites under arbitrary strain states. The Eshelby-Mori-Tanaka model was also adopted by Feng and Jiang [49] and García-Macías et al. [50,51] to simulate the piezoresistivity of CNTsreinforced composites. In those works, the piezoresistivity properties were simulated through three different mechanisms driven by mechanical strains: (i) volume expansion, (ii) filler reorientation and (iii) changes in the percolation threshold. Closed-form solutions to the formulation in Ref. [51] were later derived by Buroni and García-Macías [52] through generalized spherical harmonics series expansions, allowing to obtain analytical expressions of the effective conductivity and piezoresistive coefficients of self-sensing composites doped with rod-like inclusions. Recently, Triana-Camacho and co-authors [53] reported the development of an effective medium model to compute the impedance properties of CNTs/cement composites. While MFH offers a formal framework to relate the physical mechanisms governing the electrical transport properties of SSCCs, such modelling approaches may become overly simplistic for certain applications and have limited applicability to simulate electric time-variant systems. Numerical simulation methods instead allow to faithfully represent the microstructure of SSCCs and the interaction between their constituent phases. Numerical approaches primarily include molecular dynamics simulation (MDS), atomistic-continuum modelling, and numerical homogenization models (refer to Refs. [54,55] for a comprehensive state-of-the-art review). Despite such approaches offer high simulation accuracy, their considerable computational burden hinders their applicability to meso- and macro-scale systems. Alternatively, equivalent lumped-circuit models are being imposed as they provide a feasible and tractable framework to simulate electric time-varying systems. A noticeable example is the work by Kang et al. [56] who performed electrochemical impedance spectroscopy (EIS) testing to characterize the electrical properties of SWNT/PMMA sensors. On this basis, those authors proposed a modified Randle's circuit to represent the dynamic behaviour of the sensors. Similarly to other works, the strain sensing response was enabled by a linear relationship between the variation of the relative change of the internal electrical resistance and the axial strain. This type of approaches neglects any kind of nonlinearity in the response, thereby its applicability typically limits to small deformation ranges and filler contents sufficiently far from the percolation threshold. A similar study is the one by Loh et al. [57] on SWNT-PSS/PVA thin films, who formulated an RC-circuit model by inverse calibration from electrical impedance spectroscopy measurements. Good agreements with experimental data were obtained by defining the contact resistance, the double layer capacitance and the bulk resistance as exponentially decaying functions with time. Sanli et al. [58] proposed an RC equivalent circuit based on the impedance response of CNTs/epoxy films. An interesting aspect of that work regards the consideration of not only the dependence of the internal

resistance of the circuit on external strain, but also strain-dependent capacitive effects originating piezocapacitance. Another noteworthy contribution was made by Materazzi et al. [59] who reported an experimental campaign devoted to the assessment of the changes in the electrical resistance of SSCCs doped with CNTs under the action of sinusoidal compression loads. Their analyses showed that the amplitude of the electrical resistance variation increases with the frequency of excitation. Following that work, D'Alessandro et al. [60] proposed a Randle's equivalent circuit-based electromechanical model consisting of two resistors and a capacitor, accounting for the contact resistance (cables and electrodes), electric polarization and internal dissipation effects. An important conclusion of that work was that the dynamic response of CNTs-based SSCCs is not monochromatic, but rather contains superharmonics. However, while the presence of superharmonics was anticipated by the model, the rising amplitude of the response with increasing frequency could not be reproduced. García-Macías et al. [61] extended the circuital model proposed in [60] considering piezoresistive, piezocapacitive, and piezoelectric effects. The reported results demonstrated that the amplification of the electric resistance of CNTs-based SSCCs with the frequency of excitation under DC power supply can only be explained by a piezoelectric effect, which was further confirmed by specific experiments. Another noticeable result of that work regards the low capacitance GFs, which agrees with the experimental evidence previously reported by Han et al. [31].

As revealed by the literature review above, although significant efforts have been exerted on the characterization of piezoresistive SSCCs (e.g. refer to [27,28]), comparatively little attention has been devoted to developing and harnessing piezoelectric rGO/cement composites for strain sensing applications. Nonetheless, SSCCs relaying on the piezoresistive principle require an external power supply to conduct strain measurements, which poses a major practical limitation for civil engineering structures, often located in remote areas with no access to an electrical grid. In this light, this work presents the manufacture and characterization of highly piezoelectric rGO/SSCCs for the development of self-powered strain sensors for SHM applications. The self-powered ability of such sensors lays on the fact that they generate their own sense signal without requiring an external power supply. To this aim, a comprehensive experimental methodology is proposed to characterize the piezoelectric coupling coefficients  $g_{33}$ and  $d_{33}$ . The methodology comprises cyclic voltammetry, open circuit potential determination (OCP), and uniaxial quasi-static compression tests. The developed methodology has been applied to laboratory samples manufactured following two different filler dispersion methods. Furthermore, a novel lumped circuital model involving piezoresistive, piezocapacitive, and piezoelectric sensing principles is developed and experimentally validated. The presented results demonstrate that samples prepared by ultrasonic cleaner dispersion achieve optimal properties, with piezoelectric coefficients about 47 times greater compared to previously reported composites in the literature. Overall, the presented results evidence the potential of rGO/cement composites to develop self-powered sensors for SHM applications, for which the proposed circuital model may constitute a powerful signal processing tool allowing to translate the electrical output into strain signals. To the best of the authors' knowledge, this work presents the first evidence in the literature of the use of piezoelectric rGO/SSCCs for mechanical strain sensing applications requiring no external power supply.

The paper is organized as follows. Section 2 describes the manufacturing process and the testing details. Section 3 reviews the formulation of previously reported equivalent lumped circuits and presents the newly proposed model. Section 4 presents the numerical results and discussion and, finally, Section 5 concludes the paper.

# 2. Methods and materials

This section presents the materials, manufacturing process and testing methods adopted in this work. The manufacturing process of the



Fig. 1. Fabrication stages of rGO-based cement composites.

specimens followed two different dispersion methodologies, and the resulting dispersions were later characterized by Dynamical Light Scattering (DLS). Afterwards, the effective piezoelectric properties were characterized by cyclic electromechanical tests. Finally, this section also presents the theoretical formulation of the developed equivalent lumped circuit to replicate the electromechanical response of piezoresistive/piezocapacitive/piezoelectric rGO/cement composites.

#### 2.1. Materials and preparation of specimens

Bulk rGO powder (5–10 layer flakes with average thickness around 500  $\mu$ m) was obtained in the "*Laboratorio de Espectroscopia Atómica y Molecular*" (LEAM) of the "*Universidad Industrial de Santander*" (Colombia) following the modified Hummers' method set out by Arenas and Marcano et al. [62,63]. Other materials used in the admixtures included standard Portland cement produced under the Colombian standard NTC121 (equivalent to ASTM C1157); ultrapure water from a Milli-Q IQ 7000 equipment; and 2.0 mm copper desoldering wires used as embedded electrodes.

The fabrication process of the specimens is schematically illustrated in Fig. 1. In the first place, 0.24 mg of rGO powder is added to 80 mL of ultrapure water. Separately, two mixtures are dispersed for 2 h using an ultrasound cleaner machine model Branson 1800 (a.1) with a working frequency of 40 kHz. Simultaneously, the other two mixtures are dispersed using an ultrasonic tip set (a.2) with an energy of 490 J and amplitude of 40% applied during 30 min with (on/off) cycles of (20 s/20 s), respectively, following the same prescriptions indicated by Echeverry-Cardona et al. [64] for CNTs-based composites. The two different dispersion techniques used in this work, namely (i) ultrasonic cleaner and (ii) ultrasonic tip set, are hereafter referred to as Method 1 and Method 2, respectively. For each dispersion method, 180 mL of rGO/water solutions are prepared (b). Afterwards, Portland cement is mixed with the rGO/water solutions with a water-to-cement ratio (w/c) of 0.47, achieving rGO dispersions of 0.14% by cement mass. Note that such a filler content is far below the percolation threshold, which has been reported in the literature to be around 2.0 wt% (see e.g. [65,66]). Then, the admixture is poured into cylindrical moulds (c) which contain copper desoldering wires arranged forming two parallel plane electrodes. The electrodes are fixed into the moulds using two holes with a separation distance of 20 mm. Paper tape is used to secure the attachment between the outgoing electrodes and the mould surface. In particular, 3 cylindrical specimens (labelled with si-Mj with i = 1, ..., 3 and j = 1, 2 denoting the *i*th sample number and the *j*th dispersion method) with diameters of 30 mm and lengths of 60 mm are prepared for each dispersion method according to the ASTM C349-18 standard [67]. In this light, the repeatability of the material is statistically assessed through the analysis of variance (ANOVA) test (significance level of 0.05) of the three samples. Then the moulds are put on a vibration table for 10 min to minimize the presence of air bubbles in the samples (d). Once finished, the specimens are left to dry at room temperature for 24 h (e), followed by a curing period of 28 days in ultrapure water (specific resistance of 18 M $\Omega$ ) (f). When the curing stage is completed, the samples are further dried in an oven at 40 °C for 24 h as similarly adopted in [64] (g), after which the specimens are ready for electromechanical characterization (h).

It is important to remark that the particle size distribution of rGO is directly determined by the dispersion methodology and the sonication time as previously reported by other authors such as Kiamahalleh et al. [68]. The quality of the dispersion determines the electromechanical behaviour of the resulting composite, being in general desirable to avoid the formation of filler clusters acting as microstructural defects. The rGO dispersions obtained by the two different methodologies described above were characterized by DLS at a 90° angle using a Litesizer 500 particle analyzer from Anton Paar. To do so, the dispersions were diluted in water in a proportion of 1:100 and later poured into quartz cuvettes. The particle analyzer internally tests the dispersions in triplicate, each one obtained as the average between 6 measurements. The weighted intensity versus particle size curves obtained by dispersion Methods 1 and 2 are show in Fig. 2. It is noted in this figure that Method 1 yielded the smallest particle size with a mean value of 391 nm (intensity peak of 14.68%), considerably smaller than the one obtained by Method 2 with a mean particle size mean of 1681 nm (intensity



Fig. 2. Particle size distributions of rGO dispersed in 1:100 water for dispersion Methods 1 and 2.

peak of 14.86%). This is conceivably ascribed to the fact that, although ultrasonic tip dispersion (Method 2) generates denser energy compared to the ultrasonic cleaner (Method 1), high sonication energies may increase the charge density by the oxygen functional groups and induce the reaggregation of rGO sheets as shown in [68] and, thus, result in larger particle sizes. It is important to remark that close agreements can be found between the average particle size obtained by Method 1 in this work and previously reported results by Wang and co-authors [69].

#### 2.2. Experimental assessment of piezocapacitive/piezoelectric properties

The mechanical response of the manufactured specimens was characterized through displacement-controlled uniaxial compression tests with a universal testing machine Servosis ME 405. In parallel, the electrical behaviour of the samples was characterized using a Potentiostat PST-11 from DinTech manufacturer according to the experimental set-up shown in Fig. 3(a). The two electrodes of the specimens were connected to the reference electrode (RE) and the working electrode (WE), while the ground was connected to the grip of the universal testing machine. The data extracted from the universal testing machine and potentiostat PST-11 were synchronized in time and processed using Python scripts written in Google Colab.

A novel methodology involving cyclic voltammetry and two sets of loading conditions was developed to assess the piezocapacitive and piezoelectric properties of rGO/cement composites. Firstly, the specimens were subjected to stair-step loading with compression loads increasing from 50 kg to 250 kg, as shown in Fig. 3(c). For each step in the loading sequence, cyclic voltammetry was performed to estimate the capacitive properties of the samples and their dependence on the applied strain. Cyclic voltammetry is a popular electroanalytical technique to study the electrochemical properties of analytes. This technique consists in applying a cyclic voltage with a triangular waveform (Fig. 3 (b.1)) to the WE with respect to the RE and measuring the produced electric current (potentiostatic). The results of cyclic voltammetry are typically presented in the shape of current–potential curves (cyclic voltammograms) as sketched in Fig. 3 (b.2). In this light, the capacitance *C* of the samples can be obtained as:

$$C = \frac{1}{2s_r \Delta v} \int i(v) \,\mathrm{d}v,\tag{1}$$

where *i* is the electrical current, *v* the voltage, *s<sub>r</sub>* the scan rate, and  $\Delta v$  the voltage window, i.e., the amplitude of the applied voltage waveform. In this work, *v* spans from -0.5 V to 0.5 V, that is  $\Delta v = 1$  V. It is important to remark that the increases in the electrical current and the voltage difference increment also depend upon the

scan rate, manifesting the highest pseudocapacitance in a rGO-based composite when cyclic voltammetry is set below 50 mV/s as previously reported by Czepa and co-authors [70]. For this reason, the electrical characterization was conducted in this work with a scan rate of  $s_r = 25$  mV/s.

Afterwards, the samples were subjected to triangular cyclic loads with a fixed displacement rate of 1 mm/min (i.e. 0.05 Hz) (see Fig. 3(d)) to measure their piezoelectric response. In these tests, the correlation between the generated voltage and mechanical forces was estimated through linear regression. The estimated correlations (units of V/N) were multiplied by the conversion factor  $A/s_e$  to estimate the voltage piezoelectric coefficient  $g_{33}$ , with A and  $s_e$  being the cross-section area of the specimens and the distance between the electrodes, respectively. Note that this conversion factor allows transforming voltage into electric field and the mechanical load into stress as suggested by Al-Qaralleh [44]. Then, the correlation between the generated electric field and the corresponding stress represents the voltage piezoelectric coefficient  $g_{33}$ , which can be related to the piezoelectric charge coefficient  $d_{33}$  as follows:

$$d_{33} = C \frac{s_e}{4} g_{33}.$$
 (2)

# 3. Equivalent piezoresistive/piezocapacitive/piezoelectric lumped circuit

The main electronic transport mechanisms hypothesized in this work to describe the electromechanical behaviour of rGO-based cement composites are sketched in Fig. 4. These account for three different strain-sensing principles, namely (i) piezoresistivity, (ii) piezocapacitance, and (iii) piezoelectricity. As anticipated above, the electrical conductivity of carbon-based composites has been extensively ascribed to the coupled contribution of electron hopping and conductive networking mechanisms. In this light, (i) piezoresistivity arises as a result of three main strain-driven mechanisms: volume expansion, filler reorientation and, changes in the tunnelling resistance (see e.g. [71]). Firstly, the effects of filler reorientation and changes in the tunnelling resistance have been shown to be dominant for filler contents below the percolation threshold [51], as it is the case in the present work. Tunnelling effects arise among particles with a separation less than a certain cut-off distance (≈0.5 nm for cementitious media [72]). Therefore, as the material experiences compression, the inter-particle distance decreases, increasing the amount of charges flowing through quantum tunnelling effects and so the overall electrical conductivity. Instead, strain-induced volume expansion effects become dominant for filler contents far from the percolation threshold. Since carbonbased fillers are often orders of magnitude stiffer than the matrix, the deformation is primarily sustained by the matrix phase, which leads to variations in the apparent filler volume fraction [73]. As the volume decreases under compression, the inter-particle distance decreases rising the likelihood of forming new conductive paths and increasing the effective electrical conductivity. These phenomena explain the frequently observed piezoresistive behaviour in SSCCs in which compressive strains lead to increases in the overall electrical conductivity, and vice versa for tensile strains (see e.g. [74]). The origin of (ii) piezocapacitive effects can be intuitively explained by the inset in Fig. 4. In general, there exists a dielectric gap between every two conductive particles which can be conceived as a RC system with  $R_{gap}$  and  $C_{gap}$  corresponding to the contact resistance and capacitance, respectively. Under the action of a compression load, the distance between the particles is on average reduced, which will result in a decrease of  $R_{\rm gap}$  and an increase in the  $C_{\rm gap}$  value. Hence, in general, compression loads will lead to decreases in the electrical resistance and decreases in the overall capacitance. Finally, as sketched in Fig. 4, the origin of (iii) piezoelectric effects in rGO/cement composites is ascribed to the polarization of the sets formed by rGO sheets and cement hydration products such as calcium silicates [75].



Fig. 3. Experimental setup to characterize the electromechanical properties of rGO-based cement composites (a), and schematic representation of cyclic voltammetry (b). Loading conditions: stair-step (c) and triangular cyclic loading (d).



Fig. 4. Schematic representation of the physical phenomena governing the electric transport properties of rGO/cement composites: (i) piezoresistivity, (ii) piezocapacitance, and (iii) piezoelectricity.

The most commonly adopted circuital model to describe the electrical behaviour of SSCCs is the modified Randle's circuit depicted in Fig. 5(a). This model represents the active volume of the specimen (portion of the material in between the electrodes) as a capacitor  $C_m$  and a resistor  $R_m$  in parallel. In this light, piezoresistivity and piezocapacitance can be simulated through the strain-dependent definition of  $R_m$  and  $C_m$ , respectively. Generally, the relationship between the relative change in the internal electrical resistance and the axial mechanical strain  $\epsilon(t)$  (negative in compression) is assumed to be linear:

$$\lambda_{R_m} = \frac{\frac{\Delta R_m}{R_m^2}}{\varepsilon(t)},\tag{3}$$

where  $\lambda_{R_m}$  denotes the gauge factor of the sensor,  $R_m^o$  is the unstrained internal resistance, and *t* denotes the time variable. Similarly, piezoca-pacitive effects can be simulated through a second gauge factor,  $\lambda_{C_m}$ , as [58]:

$$\lambda_{C_m} = -\frac{\frac{\lambda C_m}{C_m^o}}{\varepsilon(t)},\tag{4}$$

10



Fig. 5. Piezoresistive/piezocapacitive unit sensing cell (a). Piezoresistive/Piezocapacitive/Piezocapacitive/Piezoelectric sensing cell (b). Proposed equivalent lumped circuit model to simulate the electrical response of piezoresistive/piezocapacitive/piezoelectric rGO-based cement composites (c).

with  $C_m^o$  denoting the unstrained capacitance.

The contribution of piezoelectric effects can be incorporated in the shape of a strain-dependent current source as sketched in Fig. 5(b) as previously proposed by García-Macías *et al.* [61]. Assuming the linear piezoelectricity theory holds, the electric current i(t) introduced in the circuit is directly related to the time derivative of the applied strain as:

$$i(t) = S_q \frac{\mathrm{d}\epsilon(t)}{\mathrm{d}t},\tag{5}$$

with  $S_q$  being a linear piezoelectricity coefficient. A similar relationship has been also suggested by several authors for other cement-based composites (see e.g. [76]). Nevertheless, the electric current measurements obtained from the rGO/cement composites manufactured in this work did not exhibit a clear linear relationship with the applied strain. These results, as shown later in Section 4.2, may indicate that linear piezoelectricity theory may not apply to this material.

From the previous discussion, a novel lumped circuit is proposed in this work to simulate the electromechanical response of rGO-based cement composites. The model, shown in Fig. 5(c), was inspired by the Butterworth-Van Dike circuit, which is commonly used to describe the piezoelectric effects in ceramics (see e.g. [77]). The proposed model accounts for three different branches in parallel representing the piezoelectric, piezoresistive, and piezocapacitive contributions. The piezoelectric branch is defined by a general current source i(t) with no assumptions upon its dependency with the applied mechanical strain. Discussion on the origin and definition of this term are reported hereafter in Section 4. The piezoresistive branch is represented by a linear variable resistance  $R_m(\varepsilon)$  in series with an inductor *L*. The latter allows reproducing the electrical damping effects arising as a result of the appearance of current loops among the rGO sheets. Finally, the piezocapacitive branch incorporates a linear variable capacitor  $C_m(\epsilon)$ and a resistor  $R_{ct}$  representing the variable internal capacitance and the charge transfer resistance, respectively.

Let us denote with  $V_i(t)$  the voltage drops across the *i*th element in the circuit model in Fig. 5(c), with sub-index *i* indicating the particular element. In this light, making use of Kirchhoff's voltage law, that is  $V_{R_m}(t) + V_L(t) = V_{R_{cl}}(t) + V_{C_m}(t)$ , it is possible to extract the governing differential equation of the proposed lumped circuit as:

$$R_m(\varepsilon)\frac{\mathrm{d}i_L(t)}{\mathrm{d}t} + \frac{\mathrm{d}R_m(\varepsilon)}{\mathrm{d}t}i_L(t) + L\frac{\mathrm{d}^2i_L(t)}{\mathrm{d}t^2} = R_{ct}\frac{\mathrm{d}i_C(t)}{\mathrm{d}t} + \frac{i_C(t)}{C_m(\varepsilon)},\tag{6}$$

with  $i_L(t)$  and  $i_C(t)$  denoting the electric currents crossing the inductor and the piezocapacitive branch, respectively. Then, combining Eq. (6) with the Kirchhoff's current law  $i(t) = i_C(t) + i_L(t)$ , the differential equation in terms of current  $i_C(t)$  can be obtained as follows:

$$L\frac{d^{2}i_{C}(t)}{dt^{2}} + \left(R_{m}(\varepsilon) + R_{ct}\right)\frac{di_{C}(t)}{dt} + \left(\frac{dR_{m}(\varepsilon)}{dt} + \frac{1}{C_{m}(\varepsilon)}\right)i_{C}(t)$$
$$= L\frac{d^{2}i(t)}{dt^{2}} + R_{m}\frac{di(t)}{dt} + \frac{dR_{m}(\varepsilon)}{dt}i(t).$$
(7)

Introducing the strain dependency of  $R_m$  and  $C_m$  from Eqs. (3) and (4), Eq. (7) can be rewritten after some manipulation as:

$$L\frac{\mathrm{d}^{2}i_{C}(t)}{\mathrm{d}t^{2}} + \left[R_{m}^{o}\left(1+\lambda_{R_{m}}\varepsilon(t)\right)+R_{ct}\right]\frac{\mathrm{d}i_{C}(t)}{\mathrm{d}t} + \left[R_{m}^{o}\lambda_{R_{m}}\frac{\mathrm{d}\varepsilon(t)}{\mathrm{d}t}\right] + \left[R_{m}^{o}\lambda_{R_{m}}\frac{\mathrm{d}\varepsilon(t)}{\mathrm{d}t}\right]^{-1}i_{C}(t) = I(t), \tag{8}$$

with

$$I(t) = L \frac{\mathrm{d}^2 i(t)}{\mathrm{d}t^2} + R_m^o \left(1 + \lambda_{R_m} \varepsilon(t)\right) \frac{\mathrm{d}i(t)}{\mathrm{d}t} + R_m^o \lambda_{R_m} \frac{\mathrm{d}\varepsilon(t)}{\mathrm{d}t} i(t).$$
(9)

Note that Eq. (8) is a second-order ordinary differential equation with variable coefficients that needs to be solved numerically. In this work, the proposed circuit has been implemented in Matlab Simulink environment, and the Runge–Kutta integration method has been used to extract numerical solutions to Eq. (8). Once the capacitive current  $i_C(t)$  is obtained, the voltage generated by the sensor can be computed as:

$$V(t) = C_m^o \left( 1 + \lambda_{C_m} \varepsilon(t) \right) \frac{\mathrm{d}i_C(t)}{\mathrm{d}t} + R_{ct} i_C(t).$$
(10)

# 4. Results and discussion

The estimation of the piezoelectric coefficients  $g_{33}$  and  $d_{33}$  through cyclic voltammetry is firstly presented in Section 4.1. Afterwards, Section 4.2 reports the experimental characterization of the electromechanical response of the manufactured rGO/cement composites under cyclic quasi-static compressive loading. Finally, Section 4.3 reports the calibration of the lumped equivalent circuital model previously introduced in Section 3.

# 4.1. Piezoelectric coefficients

Following the mechanical load history described in Fig. 3(c), Figs. 6 (a) and (b) report the output voltages versus the applied compression for dispersion Methods 1 and 2, respectively. It is first noted from these results that specimens fabricated by Method 1 (Fig. 6(a)) generated the highest voltages, around three times larger than those generated by samples fabricated following Method 2 (Fig. 6(b)). This is ascribed to the poorer filler dispersion obtained by Method 2 as previously reported in Section 2.1. The presence of rGO aggregates represents microstructural defects hampering the transportation of mobile ions in the composite, which affects the effective piezoelectric response. This effect is possibly compounded by the deterioration of the composite's effective mechanical properties as previously reported by Kiamahalleh et al. [68], who highlighted that rGO aggregates compromises the compatibility between the fillers and the host structure (cement). As a



Fig. 6. Piezoelectric voltage versus compression load for rGO-based cement composites with rGO dispersed by (a) ultrasonic cleaner (Method 1) and (b) ultrasonic tip set (Method 2).



Fig. 7. Cyclic voltammograms (scan rate: 25 mV/s) from rGO/cement specimen s1-M1 under increasing compression loads P.

result, the filler-matrix load transfer capability is reduced, presumably affecting the piezoelectric properties of the composite as well. It is interesting to note in Fig. 6 that, despite the specimens were electrically discharged before testing, there exist some initial voltages indicating the presence of transferences of electric charges between the specimens and the environment. Therefore, we will concentrate on the linear voltage change with respect to the mechanical load. On this basis, the voltage piezoelectric coefficients  $g_{33}$  have been extracted from the slopes of these curves as previously reported in Section 2.2, and the estimated coefficients are collected in Table 1.

With the aim of computing the piezoelectric charge coefficients  $d_{33}$  according to Eq. (2), cyclic voltammetry was conducted for all the different load steps in the loading sequence depicted in Fig. 3(c). As an illustrative example, the voltammograms obtained for s1-M1 are shown in Fig. 7. In this figure, the sudden rises in the electric current following the lowest peaks in the voltammograms ( $\approx$ -0.26 V) indicate the presence of reduction-oxidation (redox) reactions through which rGO particles are reduced on the electrodes. On the other hand, the highest electric current peaks are related to the graphene oxidation process [78], which occurs at  $\approx 0.25$  V. This is conceivably explained by the high content of oxygen in the rGO groups, which accelerates the redox process at the cement-rGO-electrode interfaces. Furthermore, when the specimen is subjected to increasing compressive loads, the voltage difference between the current peaks experiences slight increases. Specifically, the output voltage related to the electric current peak stemming from the graphene oxidation raises from 0.4 V (unloaded) to 0.51 V (compression load of 2.5 kN). This fact evidences that the capacitive component grows along with the strain-induced increment of the electrical conductivity of the rGO/cement composite.

The capacitances of the specimens were computed as functions of the exerted mechanical loads as previously indicated in Section 2.2. To do so, the area under the voltammograms was calculated for every compression step according to Eq. (1). The capacitance versus compression curve obtained from the voltammograms in Fig. 7 is shown as an illustrative example in Fig. 8. Note in this figure that the capacitance exhibits a clear non-linear behaviour with respect to the applied mechanical load. Specifically, the capacitance experiences considerable increases from 240  $\mu$ F until 300  $\mu$ F for mechanical loads raising from 0 kN to 1 kN. Nonetheless, the capacitance achieves a saturation zone for compression loads above 1 kN. In this region, the capacitance tends to converge to a stable value ( $\approx$ 306  $\mu$ F), which indicates a limit in the creation of new electric dipoles responsible for the polarization properties of the rGO/cement composite.

The exact same procedure was applied to all the manufactured specimens, and the resulting piezoelectric charge coefficients  $d_{33}$  are reported in Table 1. For comparison purposes, the piezoelectric coefficients reported in Refs. [36,38,44] are also included in this table. Firstly, it is noted that the piezoelectric coefficients  $g_{33}$  and  $d_{33}$  are considerably larger for the specimens fabricated by Method 1 compared to those prepared with Method 2. In particular, the  $d_{33}$  coefficient for Method 1 amounts to 1122.28 ± 246.67 pC/N, a value about 2 times the coefficient obtained by Method 1, and 47 times larger that those reported in Ref. [36] for PZT/cement composites. Such a high value indicates the relative tendency of the developed rGO/cement composites to hold the electric polarization state for a long period of time. On the other hand, the voltage piezoelectric coefficients  $g_{33}$  are considerably lower than those reported for cement composites doped with piezoelectric ceramics in Refs. [36,38]. Nevertheless, note that the obtained coefficients  $g_{33}$  are only slightly lower than the one reported

#### Cement and Concrete Composites 139 (2023) 105063

#### Table 1

Piezoelectric voltage/charge coefficients,  $g_{33}$  and  $d_{33}$ , and generated energy of rGO/cement samples manufactured following dispersion Methods 1 and 2. (Tolerances represent the standard deviation values obtained through ANOVA analysis of the three samples manufactured per dispersion method).

Dispersion method	<i>g</i> <sub>33</sub> [10 <sup>-5</sup> mVm/N]	d <sub>33</sub> [pC/N]	Generated energy density [nJ/cm <sup>3</sup> ]	
Method 1	$13.02 \pm 0.66$	1122.28 ± 246.67	38.65 ± 10.42	
Method 2	$5.02 \pm 1.18$	593.11 ± 147.42	$10.30 \pm 2.61$	
Cement paste	3.70 [44]	-	-	
Polarized cement (5 V)	12.09 [44]	-	-	
PVDF/PZT/cement	2570 [36]	24 [36]	-	
CNTs/PZT/rGO/cement	6000 [38]	-	-	



Fig. 8. Capacitance versus compression load for rGO/cement specimen s1-M1 obtained through cyclic voltammetry.

by Al-Qaralleh [44] for cement paste polarized while curing under an external voltage of 5 V.

It is important to mention that previous research has reported the potential effects of working environmental conditions on the piezoresistive properties of SSCCs (e.g. refer to [79]). Specifically, as mentioned in the introduction, the work by Dong et al. [24] highlighted the need to select an adequate water content to achieve optimal sensitivity in CB/cement sensors. However, the analysis of such effects upon the intrinsic piezoelectric properties of SSCCs has yet to be fully addressed, most works in the literature focusing on piezoceramic-cement composites. It is worth noting the work by Wittinanon and co-authors [80], who reported on the effects of the w/c ratio on the piezoelectric properties of BZT/cement composites. Their results demonstrated that the piezoelectric coefficients increase with the w/c ratio, achieving maximum sensitivities of  $d_{33} = 16.5$  pC/N and  $g_{33} = 13.1$ E-3 Vm/N for the maximum ratio investigated of w/c = 0.10. Future research efforts should focus on analysing the influence of temperature, humidity, and rheological properties of rGO/cement piezoelectric composites to devise effective strategies to compensate their influence on the output of the sensors.

# 4.2. Electromechanical response under quasi-static compressive loading

This section reports the electromechanical characterization tests conducted under quasi-static cyclic mechanical loading as described in Fig. 3(d); simultaneously, the piezo-voltage and generated electric current were measured by the potentiostat. As an illustrative example, the time series of electrical resistance, output voltage, current and power obtained for specimen s1-M1 are reported Fig. 9. Note that, since no external power is supplied to the samples, the electric resistance is determined as the piezoelectric voltage divided by the generated electric current. In order to highlight the influence of the loading condition, the time series of mechanical strains are also included in this figure. It is noted in Fig. 9(a) that the effective electrical resistance develops inversely proportional to the applied mechanical load, which agrees with the commonly observed piezoresistive behaviour in SSCCs. In this

case, this can be conceivably explained by the coupled interaction of piezoresistivity and piezoelectricity. Note that piezoelectricity induces the appearance of electric currents, which are stored in the material due to capacitive effects and, in turn, cause increases in the apparent electrical resistance. Additionally, note that the electrical resistance exhibits increasing peaks as the loading cycles progress. Such increases in the electric resistance are not ascribed to polarization phenomena as typically observed in cementitious materials under external power supply. Instead, this electric resistance can be understood as the Thevenin resistance of the piezoelectric system. It is also interesting to note the generated voltage in Fig. 9(b) exhibits a clear rising trend with some oscillating cycles corresponding to the loading/unloading peaks. A closer inspection of the oscillations in the zoom inset in Fig. 9(b) reveals that, as the compression rises, the generated voltage increases almost linearly. Once the peak is achieved and the compression starts dropping, the piezoelectric voltage decreases non-linearly with a considerably slower rate. These decreases are possibly due to the strain-induced reduction in the amount of free electrical charge susceptible to be polarized into the material. Likewise, the available electrical charge to move across the electrodes also declines as it is consumed to produce either oxidation or reduction between the cooper electrodes and the composite.

On the other hand, it is interesting to note in Fig. 9(c) that the generated current exhibits a certain direct proportionality with the mechanical strain. This evidence contrasts with the classical theory of linear piezoelectricity, which describes the generated electrical current as directly proportional to the time derivative of the applied mechanical strain as reported above in Eq. (5). Furthermore, note that the electrical current exhibits an almost exponential decrease in time, which denotes the existence of electrical damping due to inductance effects. Instead, the time series of electrical power exhibits higher linearity with the applied strain. Specifically, note in Fig. 9(a) that, for each cycle of mechanical loading, the peaks in the time series of the electrical power increase until reaching an almost stationary value. After this, the generated electrical power exhibits clear linearity with respect to the applied strain. In this particular curve, these peaks correspond to a maximum piezoelectric-generated power of 6.8 nW (power density of 0.48 nW/cm<sup>3</sup>). Note that this value is substantially lower than the power density of 14  $\mu$ W/cm<sup>3</sup> reported by Kumar and co-authors [81] for cementitious composites doped with 1wt% PVDF and 1wt% rGO. Such a considerable difference is conceivably due to the contribution of PVDF and the large concentration of rGO used in [81], which was four times higher than the content in this work. Nonetheless, although the limited power density of the developed composites may limit their use in energy harvesting applications, the high signal-to-noise ratios in the results in Fig. 9, combined with the low power consumption of the utilized potentiostat, pave the way for the development of ultralow power-consumption sensors. This allows categorizing this class of sensors as self-powered. Finally, regarding the comparison between the samples manufactured following dispersion Methods 1 and 2, the energy density values were calculated for all the samples as reported in Table 1. It can be concluded from these results that specimens prepared following dispersion Method 1 exhibit an energy production capability 73% higher than those produced by Method 2.



Fig. 9. Electrical resistance (a), piezoelectric voltage (b), electrical current (c) and generated power (d) time series measured from specimen s1-M1 and subjected to triangular cyclic loading (frequency of 0.05 Hz).

With the aim of gaining some insight into the applicability of the developed rGO/cement composites for strain sensing applications, the electrical response of sample s1-M1 versus the applied mechanical strain is depicted in Fig. 10. In particular, the response in terms of relative variation of electrical resistance, capacitance, current, voltage, and electric power are shown in Figs. 10(a) to (d), respectively. It is clear in Fig. 10(a) that, likewise other SSCCs (see e.g. [60]), the application of increasing compression loads leads to decreases in the effective electrical resistance of the material. Nonetheless, the fractional change in terms of electrical resistance exhibits a distinct non-linear behaviour compared to previous results in the literature. Note in Fig. 10(a) that non-linearities concentrate for small strains (<2500  $\mu\epsilon$ ), while the fractional change of electrical resistance tends to exhibit a more linear behaviour for medium to moderate strain levels (>2500  $\mu\epsilon$ ). Different trends have been previously observed in the literature for cement- and polymer-based smart composites. For instance, Fu and Chung [82] reported the opposite behaviour for cement mortar samples doped with carbon fibres under DC power supplies. Specifically, their results evidenced the appearance of high linearity for low mechanical strains ( $\approx 0 - 1\%$ ), followed by increasing non-linear effects for larger strain values. In this work (no power supply is applied to the samples), instead, non-linearities are ascribed to the combined effect of piezoelectric and piezocapacitive branches as further analysed hereafter in Section 4.3. From a physical standpoint, piezoresistive effects interact with the strain-induced increases in the number of electric dipoles affecting the piezoelectric response, which conceivably gain relevance at low strain values. Specifically, the electric dipoles constrain the mobility of electrons through the piezoresistive branch so affecting the linear behaviour of the fractional change in electrical resistance. Furthermore, it is evident in Fig. 10(a) the presence of certain hysteresis effects. In particular, while the fractional change in resistance raises

until 1000% after each discharge cycle at the minimum strain value, it only increases 2% on average after the loading cycle (maximum strain value). Regarding the correlation between the output voltage and strain in Fig. 10(b), it can be seen that the piezoelectric voltage increases almost monotonically with a rate of around 6 mV/cycle. Finally, let us focus on the correlation between the applied mechanical strain and the output current and generated power in Figs. 10 (c) and (d), respectively. It is evident in these figures the existence of certain hysteresis cycles, which are comparatively larger in terms of output current. In general, these magnitudes exhibit a considerably high level of linearity with respect to the applied strain, with Pearson's correlation coefficients  $R^2$  of 0.96 and 0.89 in terms of electric power and current, respectively. The presented results suggest the use of the generated electric power as a convenient magnitude for sensing applications in SHM.

## 4.3. Lumped circuit model

This section presents the experimental validation of the equivalent lumped circuit previously introduced in Section 3. In these analyses, the main parameters of the model  $\{R_m, L, C_m, R_{ct}, \lambda_{R_m}, \lambda_{C_m}\}$  are estimated through inverse calibration using the experimentally assessed piezoelectric voltage curves presented hitherto. Specifically, the model calibration is set up as a minimization problem with an objective function or cost function involving the mean squared errors between the model predictions and the experimental data. The optimization problem is solved using the gradient-descent optimization algorithm implemented in Matlab Simulink environment. With the aim of minimizing ill-conditioning in the inverse problem and obtaining physically meaningful solutions, the model parameters are firstly bounded to physically realistic intervals. In particular, gauge factors



Fig. 10. Fractional change in resistance (a), piezoelectric voltage (b), electric current (c) and generated power (d) versus mechanical strain curves measured from specimen s1-M1 and subjected to triangular cyclic loading (frequency of 0.05 Hz).

 $\lambda_{R_m}$  and  $\lambda_{C_m}$  are constrained to the intervals  $(0, \infty)$  and  $(-\infty, 0)$ , respectively. Regarding the intervals of the passive components, resistances were constrained to  $(R_m > R_{ct})$ , while the capacitance  $(C_m)$  and the inductance (L) we forced to have positive values. In these analyses, given the previously reported suspicion on the applicability of the linear piezoelectricity theory, the electric current i(t) in Eq. (8) is directly taken from the experimental data. It is important to remark that good fittings were obtained by defining the piezoelectric current as linearly proportional to the applied mechanical strain. Nevertheless, the verification of the physical significance of such a relationship would deserve further investigation in a separate contribution, so no specific functional relationship between the piezoelectric current and the applied strain was imposed herein. The resulting model parameters for the samples fabricated following dispersion Methods 1 and 2 are reported in Table 2. Overall, it is noted that no significant differences ( $\approx 2-40\%$ ) were found between the values of  $\lambda_{R_m}$ ,  $\lambda_{C_m}$ , and  $C_m$  for dispersion Methods 1 and 2, while largest differences concentrate in the values of L,  $R_m$ , and  $R_{cl}$ . Note that the gauge factors in terms of electrical resistance  $\lambda_{R_{m}}$  are considerably low, although in the same order of magnitude as those reported by Guo et al. [26] for GO/cement composites (0.05-0.15 wt%). Such low values are ascribed to the low rGO concentration used in this work, thereby the contribution of conductive networking is presumably limited and so the contribution of piezoresistive effects. It is also interesting to note in this table that the largest values of the cement matrix resistance  $R_m$  are found for samples fabricated following dispersion Method 1 (27% higher that those fabricated using Method 2), even though the dispersion of these samples was significantly better as previously shown in Fig. 2. This can be attributed to the low filler content adopted in this work, which, being well below the percolation threshold, causes the conductive networking mechanism to become ineffective. Therefore, the appearance of filler agglomerates may be

more effective to increase the electrical conductivity as filler clusters favour the transfer of electric charges through quantum tunnelling effects. Finally, note in Table 2 that low inductance values L in the order of  $10^{-7}$  H are obtained for both methods. These values are considerably lower that those reported in other works in the literature on cement-based piezoelectric ceramic composites (see e.g. [83]). Nonetheless, the definition of the piezoelectric current directly from the experiments may limit the effective contribution of L. Thereby future further investigations are needed to evaluate the relationship between the piezoelectric current and the mechanical strain and, subsequently, the actual role of L in reproducing electrical damping phenomena.

Figs. 11 (a) and (b) show the comparison between the experimental data and the predictions of the fitted circuital model in terms of output voltage for samples s1-M1 and s1-M2, respectively. Overall, it can be concluded from Fig. 11 that the proposed circuital model can accurately represent the experimental data, including the two different regimes in the voltage curves previously discussed in Section 4.2. The model shows that the charge transfer resistance  $R_{ct}$  dominates the piezoelectric voltage as it is evident from Eq. (10). Indeed, to replicate the larger voltage levels outputted by samples fabricated following dispersion Method 1 (in Fig. 11(a) the output voltage for s1-M1 is around four times the output voltage measured in s1-M2), the model calibration determined on average a  $R_{ct}$  value 44% (10.97 k $\Omega$ ) larger than the one for dispersion Method 2 (6.10 k $\Omega$ ). Similarly close fittings were also obtained in terms of electrical resistance and electric power as shown in Figs. 12 (a) and (b), respectively. It is interesting to note in Fig. 12(a) that the circuital model can correctly describe the strong non-linear behaviour in the electrical resistance even though the linear piezoresistivity relation from Eq. (3) is adopted. Note that the reported electrical current represents the Thevenin resistance or effective resistance interconnected in parallel with the piezoelectric current

#### D.A. Triana-Camacho et al.

#### Table 2

Fitted parameters of the lumped circuit model from dispersion Methods 1 and 2 (Tolerances represent the standard deviation values obtained through ANOVA analysis of the three samples manufactured per dispersion method).

Dispersion method	L [μH]	$R_m$ [k $\Omega$ ]	λ <sub>Rm</sub> [-]	C <sub>m</sub> [μF]	λ <sub>Cm</sub> [-]	$R_{ct}$ [kΩ] [10 <sup>-2</sup> ]
Method 1	$0.15 \pm 0.00$	2899.10 ± 205.89	$1.59 \pm 0.31$	307.93 ± 80.49	$-4.96 \pm 2.17$	$10.97 \pm 2.08$
Method 2	$0.07~\pm~0.02$	$2116 \pm 67.85$	$2.21 \pm 0.40$	$416.19 \pm 36.83$	$-5.09 \pm 1.48$	$6.10 \pm 1.36$



Fig. 11. Comparison between experimental and analytical predictions of piezoelectric voltage time series. Samples s1 fabricated by dispersion Methods 1 (a) and 2 (b).



Fig. 12. Fractional change in resistance (a) and electric power (d) versus mechanical strain under triangular cyclic loading (frequency of 0.05 Hz). (Sample s1-M1).

source i(t). Therefore, these results strengthen the previous discussion on the origin of such non-linearities from the coupling between the piezoelectric/piezocapacitive/piezoresistive effects instead of a nonlinear piezoresistive behaviour. Overall, it can be concluded that the presented lumped circuit model can accurately reproduce the output voltage, electrical resistance and generated power of rGO/cement composites, including the most significant non-linearities observed during the experimental campaign.

# 5. Concluding remarks

This work has presented the preparation and characterization of highly piezoelectric rGO-based SSCCs for self-powered strain monitoring applications. To this aim, a novel comprehensive methodology involving cyclic voltammetry and quasi-static compression testing has been proposed to characterize the piezoelectric coupling coefficients. The developed methodology has been applied to laboratory samples manufactured following two different filler dispersion methods, namely through ultrasonic cleaner and ultrasonic tip. Furthermore, a novel lumped circuital model has been proposed to replicate the electromechanical response of rGO/cement composites under cyclic compressive loads. The circuital model combines a piezoelectric current source in parallel with a RL series circuit accounting for piezocapacitive and piezoresistive effects. The key findings of this work can be summarized as follows:

- The presented results and discussion have evidenced the role of the dispersion methodology of rGO/cement composites on the effective piezoelectric properties. It has been concluded that filler dispersion through ultrasonic cleaner produces more uniform rGO dispersions which, in turn, leads to higher piezoelectric voltage and energy generation capabilities.
- A novel methodology combining cyclic voltammetry measurements and mechanical testing has been developed to characterize the piezoelectric coefficients *d*<sub>33</sub> and *g*<sub>33</sub>. The presented results have evidenced piezoelectric charge coefficients *d*<sub>33</sub> about 47 times larger than previously reported results in the literature for PZT-based cement composites.
- The presented experimental results have evidenced that, given the low filler content adopted in this work, the piezoelectric response plays a leading role in the strain-sensing capability of the developed composites. Specifically, the generated electrical power has been found as a candidate magnitude for conducting self-powered strain-sensing applications given its high linearity with the mechanical strain.

• The proposed circuital model has proved capable of reproducing the electromechanical response of the developed rGO/based composites in terms of piezoelectric voltage, electrical resistance, and electric power.

The proposed lumped circuit is envisioned to provide a useful tool for tuning the fabrication of rGO/based cement composites for selfpowered SHM applications. Future research efforts shall address a further analysis of the correlation between piezoelectric current and externally applied mechanical strains. Furthermore, future investigation should cover the consideration of different rGO volume fractions covering the regimes below and above the percolation threshold, as well as different dynamic loading conditions to further validate the accuracy of the proposed circuital model.

# Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Data availability

Data will be made available on request.

## Acknowledgements

The authors would like to acknowledge the financial support from the Vicerrectoría de Investigación y Extensión at the Universidad Industrial de Santander for the grant project number 3711, entitled Dispositivo electrónico para caracterizar piezoeléctricamente cilindros de pasta de cemento, which was contributed to the fabrication of a prototype for monitoring the sensors. Daniel A. Triana-Camacho want to give special thanks to scholarships program of MinCiencias Programa de Becas de Excelencia Doctoral del Bicentenario - Corte 2, 2019. E. García-Macías was supported by the Consejería de Transformación Económica, Industria, Conocimiento y Universidades de la Junta de Andalucía (Spain) through the research project P18-RT-3128. The support and assistance of Prof. Felipe García from the University of Málaga (Spain) is also warmly acknowledged. Funding for open access charge: Universidad de Granada/CBUA.

#### References

- P.J.M. Monteiro, S.A. Miller, A. Horvath, Towards sustainable concrete, Nat. Mater 16 (7) (2017) 698–699, http://dx.doi.org/10.1038/nmat4930.
- [2] J. Lehne, F. Preston, Making concrete change, Innovation in Low-Carbon Cement and Concrete (2018).
- [3] Q.C. Truong, C.P. El Soueidy, Y. Li, E. B. A., Probability-based maintenance modeling and planning for reinforced concrete assets subjected to chloride ingress, J. Build. Eng. 54 (2022) 104675, http://dx.doi.org/10.1016/j.jobe.2022. 104675.
- [4] G.M. Calvi, M. Moratti, G.J. O'Reilly, N. Scattarreggia, R. Monteiro, D. Malomo, P.M. Calvi, R. Pinho, Once upon a time in Italy: The tale of the Morandi Bridge, Struct. Eng. Int. 29 (2) (2019) 198–217, http://dx.doi.org/10.1080/10168664. 2018.1558033.
- [5] European Commission, Discussion paper-'State of infrastructure maintenance', 2019.
- [6] Ministero delle Infrastrutture e dei Trasporti Consiglio Superiore dei Lavori Pubblici, Linee guida per la classificazione e gestione del rischio, la valutazione della sicurezza ed il monitoraggio dei ponti esistenti, 14, (3) 2020, pp. 215–240,
- [7] E. Teomete, O.I. Kocyigit, Tensile strain sensitivity of steel fiber reinforced cement matrix composites tested by split tensile test, Constr. Build. Mater. 47 (2013) 962–968, http://dx.doi.org/10.1016/j.conbuildmat.2013.05.095.
- [8] A.D.B. Ferreira, P.R. Nóvoa, A.T. Marques, Multifunctional material systems: A state-of-the-art review, Compos. Struct. 151 (2016) 3–35, http://dx.doi.org/10. 1016/j.compstruct.2016.01.028.
- [9] O.A. Mendoza Reales, et al., Reinforcing effect of carbon nanotubes/surfactant dispersions in Portland cement pastes, Adv. Civil Eng. 2018 (2018) 2057940, http://dx.doi.org/10.1155/2018/2057940.

- [10] X. Li, M. Li, Multifunctional self-sensing and ductile cementitious materials, Cem. Concr. Res. 123 (2019) 105714, http://dx.doi.org/10.1016/j.cemconres.2019.03. 008.
- [11] B. Han, Y. Wang, S. Dong, L. Zhang, S. Ding, X. Yu, J. Ou, Smart concretes and structures: A review, J. Intell. Mater. Syst. Struct. 26 (11) (2015) 1303–1345, http://dx.doi.org/10.1177/1045389X155864.
- [12] O. Galao, F.J. Baeza, E. Zornoza, P. Garcés, Strain and damage sensing properties on multifunctional cement composites with CNF admixture, Cem. Concr. Compos. 46 (2014) 90–98, http://dx.doi.org/10.1016/j.cemconcomp.2013.11.009.
- [13] A. Downey, A. D'Alessandro, M. Baquera, E. García-Macías, D. Rolfes, F. Ubertini, S. Laflamme, R. Castro-Triguero, Damage detection, localization and quantification in conductive smart concrete structures using a resistor mesh model, Eng. Struct. 148 (2017) 924–935, http://dx.doi.org/10.1016/j.engstruct.2017.07.022.
- [14] S. Ding, Y.W. Wang, Y.Q. Ni, B. Han, Structural modal identification and health monitoring of building structures using self-sensing cementitious composites, Smart. Mater. Struct. 29 (5) (2020) 055013, http://dx.doi.org/10.1088/1361-665X/ab79b9.
- [15] S. Ding, Y. Xiang, Y.Q. Ni, V.K. Thakur, X. Wang, B. Han, J. Ou, In-situ synthesizing carbon nanotubes on cement to develop self-sensing cementitious composites for smart high-speed rail infrastructures, Nano Today 43 (2022) 101438, http://dx.doi.org/10.1016/j.nantod.2022.101438.
- [16] H.B. Birgin, E. García-Macías, A. D'Alessandro, F. Ubertini, Self-powered weigh-in-motion system combining vibration energy harvesting and self-sensing composite pavements, Constr. Build. Mater. 369 (2023) 130538, http://dx.doi. org/10.1016/j.conbuildmat.2023.130538.
- [17] W. Dong, W. Li, Y. Guo, Z. Sun, F. Qu, R. Liang, S.P. Shah, Application of intrinsic self-sensing cement-based sensor for traffic detection of human motion and vehicle speed, Constr. Build. Mater. 355 (2022) 129130, http://dx.doi.org/ 10.1016/j.conbuildmat.2022.129130.
- [18] Z. Tian, Y. Li, J. Zheng, S. Wang, A state-of-the-art on self-sensing concrete: Materials, fabrication and properties, Compos. B. Eng. 177 (2019) 107437, http://dx.doi.org/10.1016/j.compositesb.2019.107437.
- [19] T.S. Qureshi, D.K. Panesar, Nano reinforced cement paste composite with functionalized graphene and pristine graphene nanoplatelets, Compos. B. Eng. 197 (2020) 108063, http://dx.doi.org/10.1016/j.compositesb.2020.108063.
- [20] L. Chang, K. Friedrich, L. Ye, P. Toro, Evaluation and visualization of the percolating networks in multi-wall carbon nanotube/epoxy composites, J. Mater. Sci. 44 (15) (2009) 4003–4012, http://dx.doi.org/10.1007/s10853-009-3551-3.
- [21] K. Ramachandran, P. Vijayan, G. Murali, N.I. Vatin, A review on principles, theories and materials for self sensing concrete for structural applications, Mater. 15 (11) (2022) 3831, http://dx.doi.org/10.3390/ma15113831.
- [22] W. Dong, W. Li, Z. Tao, K. Wang, Piezoresistive properties of cement-based sensors: Review and perspective, Constr. Build. Mater. 203 (2019) 146–163, http://dx.doi.org/10.1016/j.conbuildmat.2019.01.081.
- [23] D. Jang, H.N. Yoon, J. Seo, B. Yang, Effects of exposure temperature on the piezoresistive sensing performances of MWCNT-embedded cementitious sensor, J. Build. Eng. 47 (2022) 103816, http://dx.doi.org/10.1016/j.jobe.2021.103816.
- [24] W. Dong, W. Li, N. Lu, F. Qu, K. Vessalas, D. Sheng, Piezoresistive behaviours of cement-based sensor with carbon black subjected to various temperature and water content, Compos. B. Eng. 178 (2019) 107488, http://dx.doi.org/10.1016/ j.compositesb.2019.107488.
- [25] J. Xu, D. Zhang, Pressure-sensitive properties of emulsion modified graphene nanoplatelets/cement composites, Cem. Concr. Compos. 84 (2017) 74–82, http: //dx.doi.org/10.1016/j.cemconcomp.2017.07.025.
- [26] R. Guo, Y. Suo, H. Xia, Y. Yang, Q. Ma, F. Yan, Study of piezoresistive behavior of smart cement filled with graphene oxide, Nanomater. 11 (1) (2021) 206, http://dx.doi.org/10.3390/nano11010206.
- [27] W. Dong, W. Li, K. Vessalas, X. He, Z. Sun, D. Sheng, Piezoresistivity deterioration of smart graphene nanoplate/cement-based sensors subjected to sulphuric acid attack, Compos. Commun. 23 (2021) 100563, http://dx.doi.org/10.1016/j. coco.2020.100563.
- [28] G. Qi, Q. Wang, R. Zhang, Z. Guo, D. Zhan, S. Liu, Effect of rGO/GNP on the electrical conductivity and piezoresistance of cement-based composite subjected to dynamic loading, Constr. Build. Mater. 368 (2023) 130340, http://dx.doi.org/ 10.1016/j.conbuildmat.2023.130340.
- [29] S. Wen, D.D.L. Chung, Electric polarization in carbon fiber-reinforced cement, Cem. Concr. Res. 31 (1) (2001) 141–147, http://dx.doi.org/10.1016/S0008-8846(00)00382-3.
- [30] J. Cao, D. Chung, Electric polarization and depolarization in cement-based materials, studied by apparent electrical resistance measurement, Cem. Concr. Res. 34 (3) (2004) 481–485, http://dx.doi.org/10.1016/j.cemconres.2003.09. 003.
- [31] B. Han, K. Zhang, X. Yu, E. Kwon, J. Ou, Electrical characteristics and pressure-sensitive response measurements of carboxyl MWNT/cement composites, Cem. Concr. Compos. 34 (6) (2012) 794–800, http://dx.doi.org/10.1016/j. cemconcomp.2012.02.012.
- [32] Y.Y. Hou, M.Q. Sun, J.Z. Chen, Electrical resistance and capacitance responses of smart ultra-high performance concrete with compressive strain by DC and AC measurements, Constr. Build. Mater. 327 (2022) 127007, http://dx.doi.org/10. 1016/j.conbuildmat.2022.127007.

- [33] H. Elahi, M. Eugeni, P. Gaudenzi, A review on mechanisms for piezoelectricbased energy harvesters, Energies 11 (7) (2018) 1850, http://dx.doi.org/10. 3390/en11071850.
- [34] M. Sun, Z. Li, X. Song, Piezoelectric effect of hardened cement paste, Cem. Concr. Compos. 26 (6) (2004) 717–720, http://dx.doi.org/10.1016/S0958-9465(03) 00104-5.
- [35] K. Shi, D.D.L. Chung, Piezoelectricity-based self-sensing of compressive and flexural stress in cement-based materials without admixture requirement and without poling, Smart. Mater. Struct. 27 (10) (2018) 105011, http://dx.doi.org/ 10.1088/1361-665X/aad87f.
- [36] N. Jaitanong, R. Yimnirun, H.R. Zeng, G.R. Li, Q.R. Yin, A. Chaipanich, Piezoelectric properties of cement based/PVDF/PZT composites, Mater. Lett. 130 (2014) 146–149, http://dx.doi.org/10.1016/j.matlet.2014.05.040.
- [37] X. Xu, Z. Wang, Study on dynamic properties of 1-3 cement-based piezoelectric composites, Constr. Build. Mater. 316 (2022) 125797, http://dx.doi.org/10. 1016/j.conbuildmat.2021.125797.
- [38] H. Gong, Y. Zhang, S. Che, Influence of carbon black on properties of PZTcement piezoelectric composites, J. Compos. Mater. 44 (23) (2010) 2747–2757, http://dx.doi.org/10.1177/0021998310371550.
- [39] S. Huang, X. Li, F. Liu, J. Chang, D. Xu, X. Cheng, Effect of carbon black on properties of 0-3 piezoelectric ceramic/cement composites, Curr. Appl. Phys. 9 (6) (2009) 1191–1194, http://dx.doi.org/10.1016/j.cap.2009.01.011.
- [40] R. Potong, R. Rianyoi, A. Ngamjarurojana, A. Chaipanich, Influence of carbon nanotubes on the performance of bismuth sodium titanate-bismuth potassium titanate-barium titanate ceramic/cement composites, Ceram. Int. 43 (2017) S75–S78, http://dx.doi.org/10.1016/j.ceramint.2017.05.225.
- [41] N. Jaitanong, S. Narksitipan, A. Ngamjarurojana, A. Chaipanich, Influence of graphene nanoplatelets on morphological and electrical properties of silica fume blended cement-piezoelectric ceramic composite, Ceram. Int. 44 (2018) S137–S140, http://dx.doi.org/10.1016/j.ceramint.2018.08.131.
- [42] B. Dong, F. Xing, Z. Li, The study of poling behavior and modeling of cementbased piezoelectric ceramic composites, Mater. Sci. Eng. A 456 (1–2) (2007) 317–322, http://dx.doi.org/10.1016/j.msea.2006.11.139.
- [43] Y.L. Yaphary, S. Hu, D. Lau, R.H.W. Lam, Piezoelectricity of Portland cement hydrates cured under the influence of electric field, in: 2016 IEEE 16th International Conference on Nanotechnology (IEEE-NANO), IEEE, 2016, pp. 911–914, http://dx.doi.org/10.1109/NANO.2016.7751344.
- [44] M. Al-Qaralleh, Piezoelectric behavior of polarized fresh cement pastes under the effect of different direct current voltages, Results Eng. 14 (2022) 100430, http://dx.doi.org/10.1016/j.rineng.2022.100430.
- [45] E. García-Macías, F. Ubertini, Mathematical modeling and simulation, in: Smart Nanoconcretes and Cement-Based Materials, Elsevier, 2020, pp. 101–156, http: //dx.doi.org/10.1016/B978-0-12-817854-6.00005-2.
- [46] T. Takeda, Y. Shindo, Y. Kuronuma, F. Narita, Modeling and characterization of the electrical conductivity of carbon nanotube-based polymer composites, Polymer 52 (17) (2011) 3852–3856, http://dx.doi.org/10.1016/j.polymer.2011. 06.046.
- [47] G.D. Seidel, D.C. Lagoudas, A micromechanics model for the electrical conductivity of nanotube-polymer nanocomposites, J. Compos. Mater. 43 (9) (2009) 917–941, http://dx.doi.org/10.1177/0021998308105124.
- [48] T. Tallman, K. Wang, An arbitrary strains carbon nanotube composite piezoresistivity model for finite element integration, Appl. Phys. Lett. 102 (1) (2013) 011909, http://dx.doi.org/10.1063/1.4774294.
- [49] C. Feng, L. Jiang, Micromechanics modeling of bi-axial stretching effects on the electrical conductivity of CNT-polymer composites, Int. J. Appl. Mech. 7 (01) (2015) 1550005, http://dx.doi.org/10.1142/S1758825115400050.
- [50] E. García-Macías, A. D'Alessandro, R. Castro-Triguero, D. Pérez-Mira, F. Ubertini, Micromechanics modeling of the electrical conductivity of carbon nanotube cement-matrix composites, Compos. B. Eng. 108 (2017) 451–469, http://dx.doi. org/10.1016/j.compositesb.2016.10.025.
- [51] E. García-Macías, R. Castro-Triguero, A. Sáez, F. Ubertini, 3D mixed micromechanics-FEM modeling of piezoresistive carbon nanotube smart concrete, Comput. Methods Appl. Mech. Engrg. 340 (2018) 396–423, http://dx.doi.org/10. 1016/j.cma.2018.05.037.
- [52] F.C. Buroni, E. García-Macías, Closed-form solutions for the piezoresistivity properties of short-fiber reinforced composites with percolation-type behavior, Carbon 184 (2021) 923–940, http://dx.doi.org/10.1016/j.carbon.2021.08.083.
- [53] D.A. Triana-Camacho, D.A. Miranda, E. García-Mací as, O.A.M. Reales, J.H. Quintero-Orozco, Effective medium electrical response model of carbon nanotubes cement-based composites, Constr. Build. Mater. 344 (2022) 128293, http://dx.doi.org/10.1016/j.conbuildmat.2022.128293.
- [54] C. Qiu, Y. Su, J. Yang, B. Chen, Q. Ouyang, D. Zhang, Structural modelling and mechanical behaviors of graphene/carbon nanotubes reinforced metal matrix composites via atomic-scale simulations: A review, Compos. Part C Open Access 4 (2021) 100120, http://dx.doi.org/10.1016/j.jcomc.2021.100120.
- [55] W. Zhao, J. Sun, Z. Huang, Three-dimensional graphene-carbon nanotube reinforced ceramics and computer simulation, Ceram. Int. 47 (24) (2021) 33941–33955, http://dx.doi.org/10.1016/j.ceramint.2021.08.304.
- [56] I. Kang, M.J. Schulz, J.H. Kim, V. Shanov, D. Shi, A carbon nanotube strain sensor for structural health monitoring, Smart. Mater. Struct. 15 (3) (2006) 737–748, http://dx.doi.org/10.1088/0964-1726/15/3/009.

- [57] K.J. Loh, J.P. Lynch, B. Shim, N. Kotov, Tailoring piezoresistive sensitivity of multilayer carbon nanotube composite strain sensors, J. Intell. Mater. Syst. Struct. 19 (7) (2008) 747–764, http://dx.doi.org/10.1177/1045389X07079872.
- [58] A. Sanli, C. Müller, O. Kanoun, C. Elibol, M.F.X. Wagner, Piezoresistive characterization of multi-walled carbon nanotube-epoxy based flexible strain sensitive films by impedance spectroscopy, Compos. Sci. Technol. 122 (2016) 18–26, http://dx.doi.org/10.1016/j.compscitech.2015.11.012.
- [59] A.L. Materazzi, F. Ubertini, A. D'Alessandro, Carbon nanotube cement-based sensors for dynamic monitoring of concrete structures, in: Proceedings of the 2013 FIB Symposium, 2013, pp. 22–24.
- [60] A. D'Alessandro, F. Ubertini, A.L. Materazzi, S. Laflamme, M. Porfiri, Electromechanical modelling of a new class of nanocomposite cement-based sensors for structural health monitoring, Struct. Health Monit. (2014) 1475921714560071, http://dx.doi.org/10.1177/1475921714560071.
- [61] E. García-Macías, A. Downey, A. D'Alessandro, R. Castro-Triguero, S. Laflamme, F. Ubertini, Enhanced lumped circuit model for smart nanocomposite cementbased sensors under dynamic compressive loading conditions, Sens. Actuator A Phys. 260 (2017) 45–57, http://dx.doi.org/10.1016/j.sna.2017.04.004.
- [62] B. Arenas, N. Gutiérrez, R. Cabanzo, E. Mejía, A methodology for synthesis of reduced graphene oxide membranes for desalination of produced water, 1159, IOP Publishing, 2019, 012005, http://dx.doi.org/10.1088/1742-6596/1159/1/ 012005,
- [63] D.C. Marcano, D.V. Kosynkin, J.M. Berlin, A. Sinitskii, Z. Sun, A. Slesarev, L.B. Alemany, W. Lu, J.M. Tour, Improved synthesis of graphene oxide, ACS Nano 4 (8) (2010) 4806–4814, http://dx.doi.org/10.1021/nn1006368.
- [64] L.M. Echeverry-Cardona, N. Álzate, E. Restrepo-Parra, R. Ospina, J.H. Quintero-Orozco, Time-stability dispersion of MWCNTs for the improvement of mechanical properties of portland cement specimens, Mater. 13 (18) (2020) http://dx.doi. org/10.3390/ma13184149.
- [65] A. Al-Dahawi, M.H. Sarwary, O. Öztürk, G. Yıldırım, A. Akın, M. Şahmaran, M. Lachemi, Electrical percolation threshold of cementitious composites possessing self-sensing functionality incorporating different carbon-based materials, Smart. Mater. Struct. 25 (10) (2016) 105005, http://dx.doi.org/10.1088/0964-1726/25/10/105005.
- [66] N. Zhang, W. She, F. Du, K. Xu, Experimental study on mechanical and functional properties of reduced graphene oxide/cement composites, Mater. 13 (13) (2020) 3015, http://dx.doi.org/10.3390/ma13133015.
- [67] ASTM International, ASTM standard c 349-18: Standard test method for compressive strength of hydraulic-cement mortars (using portions of prisms broken in flexure), 2018.
- [68] M. Valizadeh Kiamahalleh, A. Gholampour, D.N. Tran, T. Ozbakkaloglu, D. Losic, Physiochemical and mechanical properties of reduced graphene oxidecement mortar composites: Effect of reduced graphene oxide particle size, Constr. Build. Mater. 250 (2020) 118832, http://dx.doi.org/10.1016/j.conbuildmat. 2020.118832.
- [69] L. Wang, S. Zhang, D. Zheng, H. Yang, H. Cui, W. Tang, D. Li, Effect of graphene oxide (GO) on the morphology and microstructure of cement hydration products, Nanomater. 7 (12) (2017) 429, http://dx.doi.org/10.3390/nano7120429.
- [70] W. Czepa, S. Witomska, A. Ciesielski, P. Samorì, Reduced graphene oxidesilsesquioxane hybrid as a novel supercapacitor electrode, Nanoscale 12 (36) (2020) 18733–18741, http://dx.doi.org/10.1039/D0NR05226D.
- [71] E. García-Macías, A. D'Alessandro, R. Castro-Triguero, D. Pérez-Mira, F. Ubertini, Micromechanics modeling of the uniaxial strain-sensing property of carbon nanotube cement-matrix composites for SHM applications, Compos. Struct. 163 (2017) 195–215, http://dx.doi.org/10.1016/j.compstruct.2016.12.014.
- [72] J. Xu, W. Zhong, W. Yao, Modeling of conductivity in carbon fiber-reinforced cement-based composite, J. Mater. Sci. 45 (13) (2010) 3538–3546, http://dx. doi.org/10.1007/s10853-010-4396-5.
- [73] C. Feng, L. Jiang, Investigation of uniaxial stretching effects on the electrical conductivity of CNT-polymer nanocomposites, J. Phys. D Appl. Phys. 47 (40) (2014) 405103, http://dx.doi.org/10.1088/0022-3727/47/40/405103.
- [74] A.I. Madbouly, M.M. Mokhtar, M. Morsy, Evaluating the performance of rGO/cement composites for SHM applications, Constr. Build. Mater. 250 (2020) 118841, http://dx.doi.org/10.1016/j.conbuildmat.2020.118841.
- [75] S. Lv, Y. Ma, C. Qiu, T. Sun, J. Liu, Q. Zhou, Effect of graphene oxide nanosheets of microstructure and mechanical properties of cement composites, Constr. Build. Mater. 49 (2013) 121–127, http://dx.doi.org/10.1016/j.conbuildmat.2013.08. 022.
- [76] C. Shuai, Z. Zeng, Y. Yang, F. Qi, S. Peng, W. Yang, C. He, G. Wang, G. Qian, Graphene oxide assists polyvinylidene fluoride scaffold to reconstruct electrical microenvironment of bone tissue, Mater. Des. 190 (2020) 108564, http://dx.doi.org/10.1016/j.matdes.2020.108564.
- [77] J. Kim, B.L. Grisso, J.K. Kim, D.S. Ha, D.J. Inman, Electrical modeling of piezoelectric ceramics for analysis and evaluation of sensory systems, in: 2008 IEEE Sensors Applications Symposium, 2008, pp. 122–127, http://dx.doi.org/10. 1109/SAS13374.2008.4472956.
- [78] J. Kauppila, P. Kunnas, P. Damlin, A. Viinikanoja, C. Kvarnström, Electrochemical reduction of graphene oxide films in aqueous and organic solutions, Electrochim. Acta 89 (2013) 84–89, http://dx.doi.org/10.1016/j.electacta.2012.10.153.

- [79] B. Han, X. Yu, J. Ou, Self-Sensing Concrete in Smart Structures, Butterworth-Heinemann, 2014.
- [80] T. Wittinanon, R. Rianyoi, R. Potong, H.H. Pan, A. Chaipanich, Effect of water to cement ratio on acoustic impedance, microstructure, and piezoelectric properties of 0-3 barium zirconate titanate-portland cement composite with epoxy resin addition, Mater. Res. Bull. 158 (2023) 112078, http://dx.doi.org/10.1016/j. materresbull.2022.112078.
- [81] C. Kumar, A. Gaur, S.K. Rai, P. Maiti, Piezo devices using poly(vinylidene fluoride)/reduced graphene oxide hybrid for energy harvesting, Nano-Struct. Nano-Objects. 12 (2017) 174–181, http://dx.doi.org/10.1016/j.nanoso.2017.10. 006.
- [82] X. Fu, D. Chung, Effect of curing age on the self-monitoring behavior of carbon fiber reinforced mortar, Cem. Concr. Res. 27 (9) (1997) 1313–1318, http://dx.doi.org/10.1016/S0008-8846(97)00118-X.
- [83] F. Xing, B. Dong, Z. Li, Impedance spectroscopic studies of cement-based piezoelectric ceramic composites, Compos. Sci. Technol. 68 (12) (2008) 2456–2460, http://dx.doi.org/10.1016/j.compscitech.2008.04.024.